Author's response to reviews

Title: Exome sequencing reveals a novel TTC19 mutation in an autosomal recessive spinocerebellar ataxia patient

Authors:

Hiroyuki Morino (morino@hiroshima-u.ac.jp)
Ryosuke Miyamoto (miyamoto@clin.med.tokushima-u.ac.jp)
Shizuo Ohnishi (ohnishi@tanabe-neuro-hsp.jp)
Hirofumi Maruyama (hmaru@hiroshima-u.ac.jp)
Hideshi Kawakami (hkawakam@hiroshima-u.ac.jp)

Version: 2 Date: 12 September 2013

Author's response to reviews: see over
September 12, 2013
Mr. Timothy Shipley
Executive Editor, BMC Neurology
Floor 6, 236 Gray's Inn Road
London, WC1X 8HB
UK

Dear Mr. Timothy Shipley;

Thank you for your message concerning our manuscript entitled “Exome sequencing reveals a novel $TTC19$ mutation in an autosomal recessive spinocerebellar ataxia patient”, MS: 1804922176107544.

We really appreciate your proposal and have reformatted the manuscript according to the guideline of Research Articles.

We now hope that our manuscript will be suitable for the peer review process, and look forward you to consider it favorable for publication in BMC Neurology.

Yours sincerely,

Hiroyuki Morino, MD, PhD
Department of Epidemiology, RIRBM
Hiroshima University
1-2-3 Kasumi Minami-ku
Hiroshima, 734-8553
Japan

E-mail: morino@hiroshima-u.ac.jp
TEL: +81-82-257-5849
FAX: +81-82-257-5848

Changes that we made with your suggestions are as follows:

1. We have changed the chapters according to the guideline of Research Articles. We removed Case Report and Consent sections, and added Methods, Results, and Discussion ones. We have provided the accession numbers of the nucleotide sequences.

2. In Authors’ Contributions section, we have changed to indicate the authors with the initials.

3. We have added the funding statement to Acknowledgements section.